

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Bretschger, Lucas; Zhang, Lin

Working Paper Carbon policy in a high-growth economy: The case of China

Economics Working Paper Series, No. 14/201

Provided in Cooperation with: ETH Zurich, CER-ETH – Center of Economic Research

Suggested Citation: Bretschger, Lucas; Zhang, Lin (2014) : Carbon policy in a high-growth economy: The case of China, Economics Working Paper Series, No. 14/201, ETH Zurich, CER-ETH - Center of Economic Research, Zurich, https://doi.org/10.3929/ethz-a-010185416

This Version is available at: https://hdl.handle.net/10419/171644

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



CER-ETH – Center of Economic Research at ETH Zurich

Carbon policy in a high-growth economy: The case of China

L. Bretschger and L. Zhang

Working Paper 14/201 July 2014

Economics Working Paper Series



Eidgenössische Technische Hochschule Zürich Swiss Federal Institute of Technology Zurich

Carbon policy in a high-growth economy: The case of China

Lucas Bretschger*and Lin Zhang[†]

July 15, 2014

Abstract

There is widespread concern that an international agreement on stringent climate policies will not be reached because it would imply too high costs for fast growing economies like China. To quantify these costs we develop a general equilibrium model with fully endogenous growth. The framework includes disaggregated industrial and energy sectors, endogenous innovation, and sector-specific investments. We find that the implementation of Chinese government carbon policies until 2020 causes a welfare reduction of 0.3 percent. For the long run up to 2050 we show that welfare costs of internationally coordinated emission reduction targets lie between 3 and 8 percent. Assuming faster energy technology development, stronger induced innovation, and rising energy prices in the reference case reduces welfare losses significantly. We argue that increased urbanization raises the costs of carbon policies due to altered consumption patterns.

Keywords: Carbon policy; China; Endogenous growth; Induced innovation; Urbanization.

JEP Classification: Q54, O41, O53, C68

^{*}CER-ETH Center of Economic Research at ETH Zurich, CH-8092 Zurich, Switzerland. e-mail: lbretschger@ethz.ch.

[†]CER-ETH Center of Economic Research at ETH Zurich, CH-8092 Zurich, Switzerland. e-mail: linzhang@ethz.ch.

1 Introduction

China has become the world's largest greenhouse gas emitter: it consumes around fifty percent of global coal extraction and generates eighty percent of its electricity with coal. At the same time, the economic growth of China has been unprecedentedly high, with an average annual growth rate of more than ten percent over the last twenty years.¹ Future climate policies are becoming one of the top priorities for Chinese policy makers as well as for the world community, which is seeking a new international climate agreement.²

From the perspective of applied macroeconomics, it seems rewarding to inquire into the dynamic impact of restricted input use in a high-growth economy and to derive results on the size of its effects. Notably, if climate policy requires stabilization of future carbon use, the growth rates of fossil fuel inputs in China will contrast sharply with past and current income growth rates, suggesting major welfare losses with climate policies. One may also argue that a successfully growing economy is powerful in achieving a new growth trajectory. The sky-high savings and investment rates and the associated productivity development may support the necessary transition. The nexus of energy and growth is the fundamental research issue in this field. We focus on the effects of carbon policy on the economic growth.

In this paper we develop a multi-sector endogenous growth framework, including energy inputs. We argue that the assessment of climate policies, specifically in a case like China, is only accurate when we capture economic growth in an appropriate way. We use the well-known increasing-division-of-labor framework developed by Romer (1990) as a theoretical foundation of our model. Here, endogenous innovation and capital investments increase the number of goods varieties and the stock of knowledge, which supports growth by raising productivity. We extend the original theoretical framework to a multi-sector approach with energy and foreign trade; growth of each economic sector is determined endogenously. We then

 $^{^1\}mathrm{According}$ to World Bank data, the average annual growth rate of China was 10.1% in the periods 1991-2001, 10.9% between 2001-2011.

 $^{^{2}}$ So far, China's position has been to offer emission intensity targets but no emission cuts; we will study the different targets and their effects in detail below. Furthermore, we notice that regional pilot cap-and-trade programs are now being implemented as part of the current Chinese climate policy. This signals that the Chinese government is making efforts for the absolute emission cuts.

use the framework as a basis for a computable general equilibrium model, i.e. we calibrate the model with Chinese input output data and study the macroeconomic effects of several potential scenarios for future climate policies.

Our results indicate that the implementation of the officially announced carbon policies until 2020 incurs a welfare cost of 0.3 percent and a reduction in annual consumption growth of 0.1 percentage points. In the medium term up to 2035, where we assume more stringent emission targets, welfare costs of climate policies are substantially higher but largely depend on the assumed reference growth rate. In the long run up to 2050, welfare costs of internationally coordinated emission reduction targets lie between 3 and 8 percent; the annual consumption growth rate is reduced by up to 0.4 percentage points.

We show the robustness of the results with respect to crucial assumptions. Notably, assuming a favorable technical development in the energy sector allows to cut welfare cost of carbon policy in the long run by half. Moreover, introducing energy prices increase in the reference case reduces the cost of climate policies by one third. The assumption of induced innovation has a major impact: with a lower effect than in our standard model, the cost of carbon policies raises significantly, while a high effect could even entail economic benefits of climate policies in the long run. In the same way, we confirm that the chosen discount rate has a major impact on the results. We also show that increasing urbanization will lead to slightly higher costs of carbon policies.

The paper relates to the literature in three aspects. First, it contributes to the emerging strand of literature on the integration of the *natural environment* into *endogenous growth* theory. Acemoglu et al. (2012) show that the effects and the optimal timing of environmental policy in an innovation-driven growth framework with directed technical change depend on the degrees of substitution between clean and dirty inputs. The effects of carbon policies in our multi-sector approach also rely on inter-sectoral substitution, but contrary to most directed technical change models, we assume economy-wide and not purely sector-specific knowledge spillovers.³ Bretschger and Smulders (2012) in their theoretical model derive that in a multi-sector economy increasing energy prices do not prevent an economy

³Investments are also targeted at specific sectors in our model but we argue it is more general to assume that sector specific improvements also build on improvements in other sectors through learning effects.

from having positive innovation and growth even under the conditions of poor input substitution. In a similar way, the present model implements poor input substitution in most sectors of the economy. Popp (2002) empirically estimates the effects of energy prices on energy-efficient innovations, concluding that both energy prices and existing knowledge have strongly positive effects on innovative activities. The effects of energy prices on investments will be especially modeled in our approach. With regard to climate policies, Gans (2012) derives that only policies directed at carbon pollution have an unambiguously positive impact on innovation. The results of Cullen (2013) suggest that subsidies for renewable energies are only rationalized by their environmental benefits if the social costs of pollution are sufficiently high. Finally, Allcott and Greenstone (2012) find limited scope for "win-win" opportunities with energy policy, i.e. possibilities to consume less energy without reducing welfare by removing existing inefficiencies. We derive from this literature a consensus that climate policies are costly but that cost depends on various factors, most importantly on the growth mechanisms and innovation. We show a concrete application of this general mechanism with the example of China.

In the field of *carbon policy assessment* the importance of China and its climate policy for global greenhouse gases stabilization is eminent.⁴ There are various recent contributions on China's climate policy using computable general equilibrium. Huebler (2011) finds that the maximum welfare loss for varying energy policies between 2020 and 2050 in China amounts to four percent. By looking at disaggregated technologies, Dai et al. (2011) argue that China has to decrease coal consumption in the electricity and manufacturing sector in order to achieve the government target of 40-45 percent emission intensity reduction by 2020. Wang et al. (2009) analyze the abatement cost of different Chinese climate policy options and show that absolute emission limits similar to the Kyoto Protocol will seriously impede the Chinese development while the impact of an 80% reduction in carbon intensity by 2050 is relatively small.⁵ As reported by Financial Times,

⁴Blanford et al. (2008) conclude that effective climate policy measures must include developing and emerging countries, especially China. Wolfram et al. (2012) explain that over the next decades nearly all of the growth in energy demand, is forecast to come from the developing world, suggesting there is likely to be a large increase in the demand for energy in the coming years.

⁵Further contributions in this context are Zhang (1998), Garbaccio et al. (1999), Liang et

Beijing's leading climate economists believe about 7.5 percent of China's GDP in 2030 is likely to be devoted to reduce emissions.⁶ These results are related to ours but the main problem with these contributions is that they are based on either static or recursive dynamic models, which do not consider inter-temporal choices. Accordingly, these approaches cannot accommodate forward-looking savings and investment behavior as definitely required by modern endogenous growth theory. To develop a fully endogenous growth model is theoretically and numerically demanding. We show the possibility of constructing such dynamic model and its application to China in the present paper.⁷

With regard to economic development in China it is generally acknowledged that pace and scale of China's economic transformation have no historical precedent, see Zhu (2012). High output growth, sustained returns on capital, and a large trade surplus are the characteristics of China's recent development, accurately studied in Song et al. (2011) using a specially constructed growth model. They state that China's economic transformation involve, not only rapid economic growth and sustainable capital accumulation but also shift on the economic structure and increased urbanization (see also the survey of Zheng and Kahn, 2013). Fisher-Vanden and Ho (2007) argue that a large share of total investment in China is invested unproductively by the government in pursuit of non-economic objectives. We conclude that we have to take sectoral development and urbanization into account when analyzing emission reduction policies in a comprehensive manner. In our model, urbanization will change consumption patterns affecting carbon emissions. Moreover, our approach employs two types of capital inputs for each sector, differing in terms of productivity.

The remainder of the paper is organized as follows. Section 2 develops the theoretical framework used for the numerical simulation model and derives the

al. (2007), and Vennemo et al. (2009).

⁶http://www.ft.com/intl/cms/s/0/cd7466e8-971f-11de-83c5-00144feabdc0.html# axzz2kzFUbs5g

⁷Other fully dynamic CGE models used to evaluate climate policies in a different context are Heggedahl and Jacobson (2011) for Norway and Bretschger et al. (2011) presenting the CITE model for Switzerland; the difference to the latter paper lies in crucial model elements like induced innovation and in the adaptation to specific issues for China like high benchmark growth, special policy targets, and special issues like the effects of urbanization. We present a largely changed version of the CITE model according to the data availability and economic structure of the country. It covers all important sectors of China's economy.

conditions for balanced growth. Section 3 describes the data and presents a calibration of the model. Section 4 presents applications of the model and the findings from the model results. Section 5 introduces urbanization to the model and analyzes the reaction of the economy when both fast urbanization and carbon policies are taking place. Finally, section 6 concludes.

2 The model framework

2.1 Overview

Figure 1 shows a diagrammatic sketch of the model. A representative, infinitely lived household supplies primary factors labor (L), research labor (L_R) , capital (K), emission permits and other inputs (V). She allocates factor incomes between consumption and investment under perfect foresight in order to maximize inter-temporal utility. Emission permits are used in fixed proportion to energy uses based on the carbon content of the different energy sources. In order to obtain the effects of endogenous growth on long-run growth clearly, the baseline model is exempt from distortions, particularly taxes or other policies, as well as spatial considerations related to urbanization. We will distinguish between rural and urban consumers in a separate section. These two types of household are differentiated in terms of consumption preferences. In order to keep the analysis simple, we do not model regionally segmented labor markets; all sectors face the same labor supply.

2.2 Growth mechanics

Based on the expansion-in-varieties mechanism in intermediate goods of Romer (1990) we construct a fully dynamic multi-sector general equilibrium model. The main purpose is to apply a theoretically rigorous growth model in an open economy with different sectors and inputs. Because high productivity gains have been characteristic of the Chinese economy we enlarge the basic model when specifying productivity. Specifically, in each sector, output Y is produced using a sector specific intermediate composite Q and composite input from other sectors B:

$$Y = \left[\alpha_Q Q^{\frac{1-\gamma}{\gamma}} + (1-\alpha_Q) B^{\frac{1-\gamma}{\gamma}}\right]^{\frac{\gamma}{1-\gamma}} \tag{1}$$



Figure 1: Overview of the model

where α_Q is a share parameter. Time indices are omitted whenever there is no ambiguity. The producers of Y goods maximize profits under perfect competition, i.e. they take prices of Q and B as given. The intermediate composite Q is manufactured based on Dixit and Stiglitz (1977) as well as on Ethier (1982), where q_j denotes the *j*th type of intermediate good and J is the total number of intermediate varieties available at a certain point of time, according to:

$$Q = \left[\int_{j=0}^{J} q_j^{\kappa} dj\right]^{\frac{1}{\kappa}} \tag{2}$$

which is the standard extension-in-varieties formulation of new growth theory with $0 < \kappa < 1$. If we assume symmetric intermediate goods, i.e. $q_j = q$, expression (2) can be simplified to:

$$Q = J^{1/\kappa - 1}X\tag{3}$$

where $X = J \cdot q$ measures aggregate input in the intermediate sector. It emerges from (3) that output Q can be raised by producing larger quantity per firm q or by increasing the number of varieties (and the number of intermediate firms) J, reflecting the gains from diversification (Dixit and Stiglitz 1977). Specifically, the term $J^{1/\kappa-1}$ measures the economies of scale on the aggregate level of the economy, because with constant X output increases in J due to increasing specialization of intermediate firms. Intermediate goods q are heterogeneous and thus incomplete substitutes among each other. Hence, each firm j providing q_j operates under monopolistic competition; the term $1/\kappa - 1 > 0$ also corresponds to the optimum markup in the intermediates' sector.

2.3 Capital accumulation

We assume each intermediate good needs one capital unit in order to be produced.⁸ Accordingly, J denotes not only the total number of varieties and firms but also the amount of capital used in the economy; $1 - \kappa$ represents the share of capital in production. With $g_J = (J_{t+1} - J_t)/J_t$ being the growth rate of capital, g_X the growth rate of intermediate production, and g_Q the growth rate of output we have from (3):

$$1 + g_Q = (1 + g_J)^{\frac{1}{\kappa} - 1} (1 + g_X)$$
(4)

On a balanced growth path, sectoral allocation in the economy is unchanged so that the output of each intermediate good remains constant, i.e. we have $g_q = 0$. Output growth (g_Q) is then solely driven by gains from specialization, expressed by $g_Q = (1+g_J)^{\frac{1}{\kappa}} - 1$. Growth is positive, provided there are positive investments $(g_J > 0)$.

Capital J is accumulated through investments I according to:

$$J_{t+1} = (1+s)[I_t + (1-\delta_t)J_t]$$
(5)

where t is the time index, s the spillover of induced innovation (see next Subsection) and δ_t the depreciation rate. Investments depend on the input of research labor L_R^9 and on other investment specific inputs, B_{inv} , according to:

$$I = \left[\xi(zJ \cdot L_R)^{\frac{\omega-1}{\omega}} + (1-\xi)B_{inv}^{\frac{\omega-1}{\omega}}\right]^{\frac{\omega}{\omega-1}}$$
(6)

⁸In Romer (1990), capital is knowledge capital in the form of blueprints. We generalize the assumption to broad capital because we want to capture not only investments into non-physical but also into physical capital in the numerical simulations below. The latter constitutes an important channel for the effects of carbon policies.

⁹This variable denotes a specific type of labor, which can be derived directly from the input/output table.

where ξ and $1 - \xi$ are share parameters, ω is the elasticity of substitution, J represents the aggregate spillover of capital size to research labor productivity, and z is the spillover intensity. More specifically, we assume that the invention of new goods varieties increases the stock of public knowledge proportionally to J which is then a free input into investment activities of the next period. Hence, the knowledge spillover zJ raises research labor productivity, counteracting decreasing returns to labor in investment activities. This common mechanism of new growth theory will be present both in the benchmark scenario and the policy applications. Accordingly, carbon policies and increasing carbon prices do not affect it.

2.4 Induced Innovation

The hypothesis of induced innovation says that an increase of the price of a specific factor is a spur to innovation increasing productivity via price-induced technical progress. The seminal empirical contribution of Popp (2002) finds strong evidence for induced innovation related to energy use. Jorgenson et al. (2013, p. 481) state that "there is massive empirical evidence of price-induced energy conservation in response to higher world energy prices beginning in 1973" which leads them to conclude that a CGE model dealing with energy necessarily needs to take this into account. Because our general spillover zJ in (6) does not change the input-output relations between reference case and policy simulation we need an additional transmission channel for increasing energy prices associated to carbon policies, which we capture with our variable s, see (5). Specifically, we assume a positive impact of energy prices on investment productivity according to: 7

$$s = \max[0, \phi \left(p_e - p_{ref} \right) / p_{ref}] \tag{7}$$

where $\phi \geq 0$ measures the impact of energy price p_e on investment productivity s and p_{ref} is the (constant) energy price in the benchmark development, so that $p_e \geq p_{ref}$. We thus assume that higher energy prices, besides having negative effects by reducing intermediate goods production (see next subsection), benefit the economy through positive learning spillovers, increasing the productivity of capital investments and leading to more efficient energy use. Of course, we will carefully calibrate ϕ and test the assumption $\phi \geq 0$ in Eqn. (7) for plausibility and robustness under different climate policy scenarios; we also discuss the case $\phi = 0$. In separate simulation we also considered the assumption of s exclusively affecting energy productivity but did not find significant changes in the results. We note that the introduction of spillovers in (5) and (6) does not create any rents which would violate the usual zero-profit conditions. However, spillovers z and s directly decrease production costs for all the firms, this also leads to accelerating the increase in the number of firms in the intermediate sector, reducing prices to eliminate rents.

2.5 Intermediate goods

Intermediate goods q_j are produced using three essential inputs: labor L, energy E, and other input V, which includes the part of capital that is not invested productively (i.e. does not accumulate like the part of capital denoted by J): ¹⁰

$$q_j = J[\varphi L_j^{\frac{\epsilon-1}{\epsilon}} + \xi E_j^{\frac{\epsilon-1}{\epsilon}} + (1 - \varphi - \xi) V_j^{\frac{\epsilon-1}{\epsilon}}]^{\frac{\epsilon}{\epsilon-1}}$$
(8)

with φ , ξ , and $1-\varphi-\xi$ being the share parameters, and ϵ the substitution elasticity between the three inputs.¹¹ By multiplying the expression by J, the production of intermediates is assumed to benefit from a knowledge spillover from capital accumulation, which means that the quantity of intermediate goods increases over time with positive investments even when the quantity of the other inputs in (8) remains constant.¹²

¹⁰The main reason we distinguish between productive and non-productive capital is that a significant part of the Chinese economy is characterized by a high degree of government regulation and state-owned firms. According to previous studies in literature we thus carefully separate total capital into the two components of accumulable and constant capital, see also Section 3.

¹¹L is different from L_R in (6) so that there is no labor reallocation between these two labor types with climate policies but our formulation of (7) captures a very similar mechanism.

¹²The assumption is necessary for the calibration of the reference case (which is a balanced growth path) but not crucial for our policy evaluations, because the effect is present both in the benchmark and with the policies.

2.6 Energy sectors

Energy (E) used for intermediate production is an aggregate of electricity E_{ele} and fossil fuels E_{fos} according to

$$E = \left[\delta E_{ele}^{\frac{1-\sigma_{egy}}{\sigma_{egy}}} + (1-\delta)E_{fos}^{\frac{1-\sigma_{egy}}{\sigma_{egy}}}\right]^{\frac{\sigma_{egy}}{1-\sigma_{egy}}}$$
(9)

where σ_{egy} is the elasticity of substitution and δ is a value share. Fossil fuels E_{fos} are further disaggregated into coal, oil, and gas, using a Cobb-Douglas function, which is omitted for the sake of brevity.

In the model, emission is a by-product of the use of energy goods for intermediate production, investment, and consumption. We assume different carbon content for various energy resources used.¹³

2.7 Sectors and trade

Below we distinguish twelve non-energy sectors (hereafter regular sectors) and four energy sectors. The output composite B in (1) reflects inter-sectoral linkages through the input-output structure of the economy. China trades with the rest of the world (aggregated into one region) on all markets for final goods Y and uses Armington demand functions to model trade, where goods of each sector are differentiated by the region where they are produced. Markets for final goods are perfectly competitive and provide goods for domestic use (D) or exports (P):

$$Y = [\alpha_d D^{1+tr} + (1 - \alpha_d) P^{1+tr}]^{\frac{tr}{1+tr}}$$
(10)

where α_d is the share of domestic use in total output Y and tr is the elasticity of substitution between D and P. There is imperfect substitution between domestically produced goods Y and imported goods M:

$$A = \left[\nu M^{\frac{\eta-1}{\eta}} + (1-\nu)Y^{\frac{\eta-1}{\eta}}\right]^{\frac{\eta}{\eta-1}}$$
(11)

where ν and $1 - \nu$ are the value shares and η is the elasticity of substitution. We assume that foreign prices are given and asset trade is disregarded in the model, so

¹³Based on the data from IPCC and China, the carbon content of gas is normalized to unity, the carbon content of coal is 1.68, oil is 1.26, and electricity 1.51 relative to gas.

that goods trade is balanced in each period.¹⁴ In each sector, the market clearing condition requires that supply equals demand.

2.8 Welfare and Consumption

Total welfare is derived from individual utilities according to:

$$W = \left[\sum_{0}^{t} \left(\frac{1}{1+\rho}\right)^{t} C^{1-\theta}\right]^{\frac{1}{1-\theta}}$$
(12)

where ρ is the utility discount rate and θ is the elasticity of inter-temporal substitution. *C* represents an aggregate of different goods, consisting of consumption of a regular sector output composite (C_y) and an energy aggregate (C_e) with an elasticity of substitution (σ_C):

$$C = \left[\zeta C_y^{\frac{\sigma_C - 1}{\sigma_C}} + (1 - \zeta) C_e^{\frac{\sigma_C - 1}{\sigma_C}}\right]^{\frac{\sigma_C}{\sigma_C - 1}}.$$
(13)

The regular sector output composite (C_y) is given by a Cobb-Douglas function, according to:

$$C_y = \prod_{ne} C_{ne}^{\beta_{ne}} \tag{14}$$

where subscript ne is a set containing twelve non-energy goods, β_{ne} shows the consumption shares of each goods respectively. We further disaggregate the energy composite into fossil aggregate and electricity consumption with an elasticity of substitution σ_{ce} :

$$C_e = \left[\iota C_{ele}^{\frac{\sigma_{ce}-1}{\sigma_{ce}}} + (1-\iota) C_{fos}^{\frac{\sigma_{ce}-1}{\sigma_{ce}}} \right]^{\frac{\sigma_{ce}}{\sigma_{ce}-1}}$$
(15)

where ι is the value share of electricity consumption in total energy aggregate and the fossil aggregate (C_{fos}) is given by

$$C_{fos} = C_{coal}^{\alpha_{coal}} C_{oil}^{\alpha_{oil}} C_{gas}^{\alpha_{gas}} \tag{16}$$

¹⁴It is true that China experiences trade surplus due to its export-oriented development strategy. If we embody this fact into our benchmark, the effect of trade surplus will exist both in benchmark and policy scenarios. When we compare the effects of carbon policy on growth to the benchmark case, such effects are marginal because most of them are canceled out. Hence, the assumption of balanced trade is not likely to affect our estimation results significantly.

where α_{coal} , α_{oil} , α_{gas} are the respective energy source consumption share in fossil aggregate with $\alpha_{coal} + \alpha_{oil} + \alpha_{gas} = 1$.¹⁵

3 Data and calibration

3.1 Input output table

The model builds on data from the Chinese input-output table (IOT) of 2010. There are good reasons to use this table: 1) it contains sufficient information on intermediate and factor inputs of the different sectors; 2) it provides information on the production structure of the four major energy sources; 3) it describes demand for non-energy and energy goods; 4) it captures necessary information on investment and R&D; 5) it distinguishes between rural and urban consumers.

We introduce twelve non-energy (regular) sectors which are agriculture (agr), mining (min), chemical industry (chm), machinery industry (mch), other industries (oin), construction (con), transport (trn), banking and financial services (bnk), private services (pse), government and public services (gse), real estate (rea), water supply industry (wat) and four energy sectors, i.e. electricity (ele), coal (coa), oil (oil), and gas (gas).

3.2 Capital share

An important issue for a dynamic model is the capital share of the economy. Fleisher et al. (2010) find that it is between 0.18 and 0.52, depending on the model specifications. Following Bai et al. (2006), Song et al. (2011) use a value of 0.5 for their numerical studies. These high capital shares are a consequence of the high savings rate but also due to the fact that all these studies do not consider energy as a separate input of production. However, in the input/output data we use it turns out that the energy share of GDP is much higher than in other (developed) economies, at about 20 percent. Furthermore, Bai et al. (2006) find evidence of misallocation of investment in China and an overestimation of capital share in statistics. To reflect the part of less productive capital, we distinguish

¹⁵The remaining equations for model closure are the standard equations on zero-profit conditions, and thus do not yield additional insights; they are available from the authors upon request.

two types of capital: non-accumulative capital and accumulative capital. Nonaccumulative capital enters the production function as an additional input besides labor and energy. Only accumulative capital contributes to the creation of new knowledge. For the accumulative capital we relate to Kuemmel et al. (2002), who estimate an average level of 0.25 for comparable growth miracles such as Japan and Lu and Zhou (2009), who estimate the cost share of capital in the periods 1978-2005 for China to fluctuate between 0.10 and 0.30. Accordingly, we assume the share of accumulative capital to be 0.25 in this study.¹⁶ Because we also include (sector-specific) non-accumulative capital the average capital share of the benchmark economy amounts to around 40 percent on average.

3.3 Other assumptions

The prices of all goods are assumed to be constant in the benchmark, which is the usual assumption for CGE models.¹⁷ We will test the impact of increasing energy prices due to increasing scarcities in a separate section below. To determine induced investment reflected in (7) we refer to the estimation of Popp (2002) who reports a long run elasticity between 0.354 and 0.421 for (energy-related) technology patents with respect to energy prices. We use a value of ϕ of 0.2 in all the sectors in the first part of the study and test for sensitivity in a separate section below.

Table 1 provides the parameter values for the different elasticities we use for numerical simulation. The values are taken from existing studies, specifically van der Werf (2007), Okagawa and Ban (2008), Hasanov (2007), and Donnelly et al. (2004). As customary in applied general equilibrium analysis we use economic value flows of the dataset to calibrate the value share and level parameters for the base year of the model. The model is calibrated to a steady-state baseline extrapolated from the base-year IOT with assumptions on growth rate of output,

¹⁶In the benchmark, the 25% applies for all the sectors, which is a necessary assumption for balanced growth. We run sensitivity check with lower and higher value of capital share, and find higher share of capital will lead to a relatively higher cost for the same emission reduction because higher capital share means lower share of energy input in production, which potentially increases the productivity of energy, and hence carbon reduction policy will raise the energy prices further relative to lower capital share case. In contrary, lower capital share declines the cost for carbon mitigation. However, such changes in cost are small.

¹⁷Specifically in the dynamic model setup, the price path over time in terms of present value in the model is calibrated to decline with a rate of 1/(1+r), where r is the interest rate.

interest rate, depreciation.

Parameter	Description	Value
γ	Elasticity of substitution between Q and	0.392 (agr)
	inputs from other sectors B	0.848 (coa, oil,
		gas, ele, chm)
		$0.518 \;({\rm mch})$
		$0.500 \;({ m min})$
		$1.264 \ (con)$
		$0.352 \ (trn)$
		0.568 (oin)
		0.492 (rest)
ε	Elasticity of substitution between the three	0.7 (agr, coa, oil,
	inputs (Energy E , labor L and other inputs V)	gas, ele, chm)
		0.8 (mch)
		0.52 (con)
		0.82 (oin)
		$0.5 (\min)$
		0.4 (rest)
ω	Elasticity of substitution between invest-	0.3
	ments in R&D (B_{inv}) and research labor L_R	
σ_C	Elasticity of substitution between energy (F)	0.5
	and non-energy goods (D) in consumption	
σ_{egy}	Elasticity of substitution between electricity	0.8
	and fossil fuels in intermediate production	
σ_{ce}	Elasticity of substitution between electricity	1.5
	and fossil fuels in consumption	
θ	Inter-temporal elasticity of substitution in	0.6
	the welfare function	
η	Trade ("Armington") elasticities	3.2 (agr)
		$4.6 \pmod{\text{mch}}$
		3.8 (oin)
		2.9 (rest)
tr	Elasticity of transformation	1
v	Elasticity of substitution between sectoral	0
	outputs for the input B	
φ	Impact of energy price on innovation	0.2

 Table 1: Parameter values for regular sectors and consumption

3.4 Time frames

We consider three different time frames for our analysis: short term (2010-2020), mid term (2010-2035), and long term (2010-2050); they differ in terms of reference growth rates and policy targets. We construct three different baseline scenarios that are designed to reflect different time frames with corresponding differences in assumed reference growth rates. The reference growth rate in the short run is assumed to be 7 percent per year, based on the 12th Five-Year-Plan report of the Chinese government, which is our reference to study the carbon policies up to 2020. Two different reference growth rates (4 percent and 7 percent) are used for the analysis of medium run scenarios with a focus on economic effects of carbon policies advocated by the International Energy Agency (IEA). In the long run, the economy is assumed to grow at an average annual rate of 4 percent in the benchmark; carbon policy targets are based on international burden sharing rules which are currently discussed for a global climate agreement. The real interest rate is assumed to be 4% following World Bank data. According to the calibration procedure, the discount rate is implicitly determined by the real interest rate, the reference growth rate of output, and the inter-temporal elasticity of substitution.

3.5 Benchmark

All policies scenarios are compared to a benchmark. In our multi-sector economy, the benchmark is assumed to be a balanced growth path meaning that all sectors grow at the same steady-state rate; i.e. there is no structural change. The population is assumed to be constant over time (a realistic assumption for China); hence, aggregate work and research labor remain unchanged. The varieties in production expand over time, entailing spillover effects and increasing productivity of intermediate goods' production. According to Equation 4, an increasing variety of goods (increasing output) can be produced with a given amount of input, which is the source of endogenous growth in our model. In the benchmark, there is no specific investment induced by changing energy prices because these are constant (i.e. s = 0). As the carbon contents of energy sources are fixed, emissions grow at the rate of energy and general output in the benchmark.

4 Implementing carbon policies

4.1 Policy scenarios

China has enacted several national and provincial energy saving regulations and codes to achieve carbon emission reductions. Prominently, the 12th of China's five-year plans for the period 2011 to 2015 aims at an emission intensity reduction of 17% in 2015 against the 2010 value. A long-term target of 40-45% emission intensity reduction in 2020 against the 2005 level has also been specified by the

government.¹⁸ These policies will be labeled CHN40 and CHN45 below. The proposal is less stringent than other countries' announcements such as the U.S. target of a 17 percent reduction of greenhouse gases¹⁹ by 2020 against 2005 because for a fast-growing economy, absolute emissions would still keep growing significantly.

China has recently evaluated the policy of emission cap from 2016 on. But according to the International Energy Agency (IEA 2010), it will likely continue to increase emissions until 2020, emitting 20% more compared to 2010. In the IEA 450 ppm scenario, which we label IEA450, (absolute) emissions in China will start to decline only after 2020. Under IEA450, total emissions in 2035 are about 71 percent of the 2010 level. The IEA450 scenario stops in 2035, but worldwide climate policies are now formulated until 2050. Hence, two scenarios with long term goals will also be considered: CER524 and CER361. These two scenarios are derived from the contribution of Bretschger (2013). He proposes a general synthetic rule for burden sharing in international climate agreement based on general equity principles, finding that an average budget per capita per year in the period 2010-2050 for China is 5.24 tons in the most favorable case (labelled CER524), and 3.61 tons in the most unfavorable case (CER361).²⁰ Considering that Chinese per capita emission in 2010 were 5.4 tons (IEA 2010), total reduction in emissions between 2010-2050 is 3 percent in CER524 and 33 percent in CER361, respectively. In addition, we study - as a reference policy scenario - a path where the emission level is kept constant at the level of 2010 until 2050 (ZERO).

Overall, we evaluate six climate policy scenarios: two government target scenario in the short run (CHN40 and CHN45); one scenario in the mid-term,

 $^{^{18}\}mathrm{As}$ in 2010, 21% emission intensity reduction has been achieved compared to the level of 2005. China needs a reduction of emission intensity by 24% and 30% in 2020 from 2010 level to reach the 40% and 45% target between 2005 and 2020 respectively.

¹⁹The target of the U.S. is based on the UNFCCC report. http://unfccc.int/meetings/ copenhagen_dec_2009/items/5264.php.

²⁰We note that these two scenarios show only what China should do disregarding the policies implemented in other countries. One way of introducing policy effects of foreign countries in our one region model framework is to vary the value of trade elasticity. If foreign countries implement less stringent carbon policy, the prices of fossil energies in China are higher than abroad. This will increase the incentives to import goods rather than producing in domestic firms. Hence, trade elasticity rise, as there is an increased preference for foreign goods. On contrary, we can decrease the value for trade elasticities to formulate the case where foreign countries implement more stringent carbon policies than China. The appendix shows such sensitivity analysis.

namely IEA450; two scenarios for the long run, CER524 and CER361; and, in addition, one scenario for all three time frames with constant emission level over time (ZERO). Technically, the emission targets are achieved by imposing carbon taxes for energy use and consumption according to the carbon content of various energy sources. Tax revenues are redistributed to consumers as a lump-sum transfer.

Figure 2 depicts the CO2 emission profiles across scenarios in China. As the economy grows at the rate of 7 percent per year, China will emit 14,278 Mt CO2 in 2020, which doubles the 2010 level (7258.5 Mt according to IEA(2010) estimation). For the two government target scenarios, the CO2 emission in 2020 is 10,852 Mt in CHN40, and 9,995 Mt in CHN45, resulting in a decline in emission by 24 percent and 30 percent respectively compared to the reference scenario. However, this is up to 50 percent increase relative to the emission level in 2010. The absolute emission keeps growing at a relatively lower rate compared to the reference scenario. These two government scenarios allow more emission than IEA450 which estimates the emission in 2020 is 9,030 Mt. CER524 and CER361 have the same emission limitation for the year 2020. Before 2020, all the five scenarios have an increase in CO2 emission, showing limited efforts for emission mitigation. The distinction happens after 2020. IEA450 requires emissions to go down to 5,164 Mt in 2035, where in CER524 6,097 Mt in 2035 is allowed, 18 percent higher than the level of IEA. The most stringent target is to keep the carbon emission at the level of 3,121 Mt after 2035 in CER361, up to 39 percent lower than the IEA level. Table 2 summarizes all scenarios implemented in this section.

4.2 Results of cost estimations

4.2.1 Short run (2010-2020)

Figure 3 shows the growth path of aggregate consumption over time across scenarios. Given the high growth rate consumption increases by a factor of two between 2010 and 2020. For the two government scenarios CHN40 and CHN45 we obtain a growth rate of around 6.91 percent against 7 percent in the benchmark. The discounted welfare loss of the two climate policies is 0.32 percent and 0.34 percent. In scenario ZERO, the welfare loss associated with keeping the emission



Figure 2: CO2 emission trajectories across scenarios

level constant is 0.84%.



Figure 3: Annual average growth rates of aggregate consumption under different time horizons

4.2.2 Medium run (2010-2035)

For the medium run we consider two reference growth rates: 7 and 4 percent. Assuming the 7 percent growth from the short run the welfare loss is 7.23 percent in the IEA450 scenario with an average growth rate of consumption at 6.57

Scenario	Time	Reference	Carbon policy
	$\mathbf{periods}$	Growth rate	Definition
Short-term			
CHN40	2010-2020	7%	40% carbon intensity reduction
CHN45	2010-2020	7%	45% carbon intensity reduction
ZERO	2010-2020	7%	Constant emission level $(5.4t \text{ per capita per year})$
Mid-term			
IEA450	2010-2035	7%	Emission profile defined in 450ppm scenario by IEA
ZERO	2010-2035	7%	Constant emission level (5.4t per capita per year)
IEA450	2010 - 2035	4%	Emission profile defined in 450ppm scenario by IEA
ZERO	2010-2035	4%	Constant emission level (5.4t per capita per year)
Long-term			
CER524	2010-2050	4%	Carbon budget of 5.24t per capita per year
CER361	2010-2050	4%	Carbon budget of 3.61t per capita per year
ZERO	2010-2050	4%	Constant emission level (5.4t per capita per year)

 Table 2: Description of scenarios in section 3.2.2

percent. As a comparison, to keep CO2 emissions constant (in ZERO) the loss in welfare amounts to 5.35 percent. Comparing these results to the short run we find two interesting issues. First, cost of carbon mitigation increases with time, even with the normal assumption of a positive discount rate. The reason is that emission cuts have higher costs with a higher income level. Specifically, in the ZERO scenario it can be seen that the cost of the policy for the first 10 years is 0.84 percent, while the cost for the next 15 years is 4.51 percent. Second, and related to that, it is not beneficial but costly to delay emissions to later periods. The reason is that an earlier redirection of inputs towards investment and growth is beneficial in our growth model. Notably, in *IEA*450, emission reduction mostly happens in the last 10 years (2025-2035) which leads to a substantial increase in welfare cost of carbon policy.

Suggesting that 7 percent annual growth up to 2035 is a too ambitious target we now reduce the reference growth rate of the economy to 4 percent in the baseline. It can be seen from Figure 3 that a lower reference growth rate makes it easier for the economy to reach the emission target, as could be expected. Consumption growth in the two policy scenarios is now 3.88 percent. Remarkably, the welfare loss in both scenarios is less than one fifth of the value with 7 percent reference growth (1.34 and 0.94 percent). Hence, even in our endogenous growth model, the cost of carbon emission mitigation increases drastically with the reference growth rate of the economy. There are two reasons for this huge difference. First, a lower GDP growth rate means lower CO2 emissions so that the differences between baseline and emission mitigation targets become smaller. Less resource and effort are required for the emission reduction with lower GDP growth. Second, in our fully fledged intertemporal approach, a lower growth rate of the economy implies a higher discount rate²¹ which reduces the present value of future cost. As the welfare loss is computed as the accumulative discounted present value over time, the estimated cost is smaller compared to the case of high growth.

With 7% reference growth					
Time frame	short-r	run (2010-	$\cdot 2020)$	medium-r	run (2010-2035)
Scenarios	CHN40	CHN45	ZERO	IEA450	ZERO
Welfar loss	-0.34%	-0.32%	-0.84%	-7.23%	-5.35%
With 4% reference growth					
Time frame	long-r	un (2010-	2050)	medium-r	run (2010-2035)
Scenarios	CER524	CER361	ZERO	IEA450	ZERO
Welfare loss	-3.10%	-8.33%	-2.57%	-1.34%	-0.94%

Table 3: Welfare loss across scenarios with different time frames

4.2.3 Long run (2010-2050)

For the long run we assume the reference growth rate of the Chinese economy being 4 percent per year. Enforcing a constant emission level over time (scenario ZERO) the growth rate of consumption becomes 3.82 percent, which is somewhat lower than the rate in the medium run. Because development is less dynamic and the time horizon is longer, we obtain a welfare loss of 2.57 percent for this policy. The first case of burden sharing from the global perspective (CER524) suggests similar results, the welfare loss amounts to 3.10 percent. CER361 showcases the most stringent climate policy scenario. The growth rate of consumption drops to 3.54 percent and welfare loss rises up to 8.33 percent, which is the highest value we obtain in the present setup.

Given the various modeling and parameter choices we made to obtain these results we have to do extensive sensitivity analysis, to which we turn next.

²¹Following the standard calibration procedure and given the real interest rate and intertemporal elasticity of substitution, the discount rate is then defined implicitly by the growth rate of output. See Rutherford (2004) and Ramer (2011) for more details.

4.3 Sensitivity analysis

There are several model issues which might be important for our results and therefore deserve closer attention. First, our model includes general learning effects but does not assume a specific technology development for the future. However, from the perspective of engineering, the development of new technologies e.g. for electricity generation from renewable energy sources as well as novel technologies for carbon capture and storage (CCS) are relevant scenarios. Second, we have assumed constant energy prices in the reference case. But as soon as increasing scarcities entail increasing energy prices (independent of any policies), the evaluation of these policies looks different. Third, varying the size of the learning effects induced by increasing energy prices affects the results. Fourth, we have to reconsider the issue of discounting in our endogenous growth model. Finally, the assumed elasticities of substitution have to be varied to see their impact on the final results.

4.3.1 Green technologies

In addition to general efficiency improvements there are two specific technical approaches to reduce greenhouse gas emissions. One is to develop and use renewable energies which are CO2 free. To promote renewable energies China has enacted its Renewable Energy Law. A specific goal for renewables is also set out in China's 12th Five Year Plan, which specifies values of 11.4 percent for total primary energy from non-fossil sources by 2015 and 15 percent by 2020; the current level is 8 percent. 20 percent of current electricity generation is attributed to renewable resources, 18 percent stems from hydropower. The other technology option is to reduce future CO2 emissions by adopting CCS. China runs one of the largest numbers of CCS pilot projects in the world. Operations of the projects include state-owned power generation, coal and oil companies.

To specify this technology evolution, we now assume a declining trend of carbon content of energy input. We consider two alternative scenarios: one showing the carbon content of electricity only declining to half of 2010's level in 2050 (labeled TC for ele), the other reflecting a declining carbon content of all energy source to half of their 2010's levels respectively (TC for all).



Figure 4: Aggregate effects comparison with green technology development in the long run

Results in Figure 4 confirm that cost for carbon mitigation can be reduced if renewables are introduced as a substitute for polluting energies. As compared to our previous results, welfare losses decrease in all three carbon policies. The reduction is much larger when technologies such as CCS can be used to decline the carbon content of fossil energies. In the most stringent scenario CER361, welfare loss of carbon policy drops from 8.33 percent to 4.5 percent, accounting for approximately half of the total welfare loss. Accordingly, aggregate consumption growth is higher than in the case without a specific technology development.

We note that the improvement of efficiency in CCS and the expansion of renewable energy in electricity generation involve additional investments which are excluded from the calculation. Hence, our estimation of the contribution of exogenous technical change to cost reduction may be overestimated.

4.3.2 Energy price effects

Based on the theory of nonrenewable resources as developed in Hotelling (1931), the optimum extraction path for non-renewable resources is one along which the resource rent increases at the rate of interest. To reflect the development path of energy prices according to the Hotelling framework for nonrenewable resources, we run separate scenarios (i.e. a series of PR scenarios with different time frames and growth rates of output) assuming that energy prices increase with the interest rate, i.e. by four percent per year in our numerical simulation.



Figure 5: Consumption over time with increasing energy price

As illustrated in Figure 5, higher energy price discourages energy consumption and increases prices of consumer goods, especially energy intensive goods. Under such conditions, consumption growth is lower than in the reference case, where energy prices are assumed to be flat. The average short run growth rate of consumption is 6.83 percent (level of reference is 7 percent). Consumption is not very sensitive to the prices in the short run but significant in the longer run, especially in the scenarios with relatively low growth rates of output.²²

To determine induced investment reflected in (7) we refer to the estimation of Popp (2002), who reports a long run elasticity between 0.354 and 0.421 for (energy-related) technology patents with respect to energy prices. As shown in Table 4, the positive effect of induced innovation alleviates the negative impact of energy price increase. The positive effect increases in the value of ϕ . So far we have assumed ϕ to be 0.2, the consequences of reducing it to 0 (no learning

 $^{^{22}}$ The welfare loss of the path with increasing energy prices is 2.2 percent relative to the reference case with constant energy prices in the short run. The loss of welfare increases to 5.5 percent with a growth rate of 6.68 percent per year in the mid-term time until 2035. If the output grows at 4 percent per year in the reference, the loss of welfare increases to 8.5 percent with a growth rate of 3.51 percent per year in aggregate consumption. The long-run effect is much stronger, aggregate consumption growth drops from 4 percent to 3.33 percent, accounting for 12.44 percent welfare loss.

spillovers from increases energy prices) and doubling to 0.4 are also given in Table 4.

Value of ϕ	Welfare change consumption					
	price effect	induced invest effect	aggregate effect	aggregate		
$\phi = 0$	-12.44%	0	-12.44%	3.33%		
$\phi = 0.2$	-12.44%	+10.31%	-2.13%	3.85%		
$\phi = 0.4$	-12.44%	+19.94%	+7.5%	4.17%		

 Table 4: Aggregate effects of increasing energy prices with induced innovation

Induced innovation (the magnitude of ϕ) has a significant impact on the effects of carbon policies. It can convert a relatively high welfare loss in the case of no induced innovation ($\phi = 0$) into a welfare gain, constituting a "win-win" situation which one might call successful "green growth". Based on empirical evidence and because we do not want to assume a value which is overly optimistic we stick with $\phi = 0.2$ in the main analysis.

In absence of induced innovation, the welfare loss of the economy is about 12.44% under the Hotelling pricing assumption. With regard to energy price development, all three long-run carbon reduction scenarios lead to an increase of energy price of more than 4 percent (our Hotelling case), meaning that carbon policies further increase the price of energy. Hence, if we think that Hotelling forces will come into play in the future, the estimated welfare losses above implicitly contain the effects due to the Hotelling price change. It is then illuminating to subtract the effects of the Hotelling energy price path from our estimations.²³ Figure 6 shows that, after separating the Hotelling energy price effect, the maximum welfare loss from CER361 declines to 6.20 percent. Put differently, the average carbon budget per capita per year in the benchmark has to be around 13.4 tons to sustain annual growth of 4 percent until 2050. The Hotelling energy price path reduces the carbon budget to 7 tons, which accounts for up to 80 percent emission reductions in the carbon policies scenarios. We conclude that in a world with increasing energy prices, emissions will be implicitly reduced through the price effects (both negative price increasing effect and positive price-induced innovation effect), and the required policy efforts to reach long-run emission targets become substantially lower, which also applies to the welfare losses.

 $^{^{23}}$ To decompose the effects of induced innovation from aggregate effect in Table 4, we assume the (Hotelling) price effect in all three scenarios is the same, the residual between price effect and aggregate effect gives the effect attributed to induced innovation.



Figure 6: Decomposition of Hotelling price effect and carbon policy effect

4.3.3 Discounting

The choice of the discount rate will affect the estimation in the long run. The model calibration for the above analysis implicitly assumes a discount rate of 1.6 percent in the model. The social planner might prefer a different discount rate to market participants and use a value of 4 percent, which is frequently used in climate policy. Using the reported consumption path from the model and together with the discount rate of 4 percent, the welfare level associated with the new discount rate can be calculated using equation 12 separately. We call this a "static approach" to welfare estimation. The higher discount rate leads to a sharp reduction of welfare loss. The cost of carbon mitigation policy in scenarios CER524, CER361, ZERO now become 0.81 percent, 3.1 percent, and 0.79 percent, respectively. The reason is that the planner values future consumption losses due to climate policy less than the households.

Instead, a "dynamic approach" analysis is conducted if we impose the higher discount rate to the individuals. The difference between the static and dynamic approach is that interest rate will be adjusted accordingly in a dynamic context. The intertemporal optimization of consumption (Keynes-Ramsey rule) suggests that the market interest rate has to rise as well, in our case from 4 percent to 6.4 percent. The reason is that the benchmark path is determined by a given growth rate. It is then still true that higher discounting reduces the cost of climate policies. But the welfare loss does not decline by a large amount compared to the original estimation since a higher interest rate makes it more expensive to invest in capital and to substitute for fading energy input. This is confirmed by the results from Table 5 which shows lower growth rates of consumption in the case of 4 percent discounting compared to the case of 1.6 percent discounting.

 Table 5: Aggregate effects comparison with different inter-temporal discounting rates

 in the long run

Scenarios	Welfare change			Aggregate c	Aggregate consumption growth		
	1.6%	4%	4%	1.6%	4%	4%	
	(benchmark)	(dynamic)	(static)	(benchmark)	(dynamic)	(static)	
CER524	-3.10%	-2.83%	-0.81%	3.79%	3.77%	3.77%	
CER361	-8.33%	-7.02%	-3.10%	3.54%	3.53%	3.53%	
ZERO	-2.57%	-2.46%	-0.79%	3.82%	3.82%	3.82%	

4.3.4 Substitution elasticities

Finally, additional sensitivity analysis on the values of the substitution elasticities are conducted to check the robustness of our results. The Appendix summarizes these results on parameter sensitivity, indicating the high reliability and robustness of our results on the cost of carbon policy.

4.4 Results of structural change

Because the model contains many sectors with important intersectoral linkages, the structural aspects of development are worth considering. In the reference case, all the sectors grow at the rate of aggregate output. But carbon policies have an impact on sectoral growth and thus change the sectoral structure of the economy. In general, energy intensive sectors tend to shrink while knowledge intensive sectors are able to grow faster.

As shown in Figure 7, climate policies affect sectoral development. In the general model, technological change and efficiency improvement stem from two substitution effects: (i) substitution between energy input and other inputs (for instance, work labor), because the price of other inputs is relatively cheaper than energy since emission cap implicitly increases the price of energy; (ii) investment in research and the spillover effects from research labor. Innovative sectors which are capital intensive in the baseline can adjust easier and alleviate the shocks from



Figure 7: Effects of carbon policies on sectoral growth in the long run

carbon tax. These two forces come from the setup of the model (see Equations 6 and 8) and can be observed from the change in the growth rates of regular sectors.

Energy intensive sectors such as Mining industry (min), machinery (mch), construction (con), transportation (trn) shrink compared to sectors such as agriculture (agr), which is labor intensive and private service sector (pse), which is capital intensive. Particularly, the min sector, as a source of primary energies, will experience a decline in production since less fossil energies are demanded in the future. It is worth noting that water supply industry (wat) declines substantially as well. This confirms the information that converting primary energy into end use energy requires a great deal of water. Hence, demand for water declines as the energy sectors shrink.

As targeted by the policy, the energy sectors suffers from the adopted policies. Within the energy sectors, two substitution effects are effective. The first is substitution between the three fossil energies. Energy sources with higher carbon content can be replaced by sources with lower carbon content since higher carbon content implies that higher tax is imposed for that energy source. It is clear from the figure that coal suffers the most. It shrinks with a rate of between -1.22 percent in ZERO and -4.15 percent in CER361, followed by oil, which still grows at a rate of 0.74 percent in ZERO but shrinks with a rate of -1.47 percent in CER361. The change of the growth rate in gas is insignificant between ZERO and CER524. To achieve the most stringent target in CER361, the production of gas has to keep almost at current level. The dependency on natural gas will have to increase since it is relatively cleaner energy source compared to others. The second effect is substitution between electricity and fossil energies. Acceleration of electrification makes it relatively easier to substitute. Specifically in China, substantial investments in power plants and grids construction enlarged the penetration rate of electricity distribution and electric equipment. However, most of the power plants in China are still coal-fired, which means electricity is carbon intensive relative to, for example, gas. This can lead to an "inverse" substitution between electricity is much lower than two of the three fossil energies in all scenarios. The decline in electricity growth is large. It drops to -0.55 percent in ZERO and -2.68 percent in CER361, which provides evidence that the second inverse substitution effect is dominant.

Figure 8 offers an overview of the change in energy mix over time across scenarios. *ZERO* shows a clearly rising share of gas and oil, and a substantially decline in the share of coal and electricity. This result illustrates an induced transition towards cleaner energy sources when climate policy is binding. The sub-figure in the bottom right illustrates the correlation between energy consumption and CO2 emission. With the emission reduction, total energy consumption also declines.

5 Urbanization and sectoral change

As predicted by the United Nations (2013), by 2050 total population in China will remain almost at today's level²⁴, even though it first increases to peak around 2025 and declines afterwards. The most dominant demographic effect is urbanization. The rate of urbanization is seen as a sign of success of economic achievement. China's population urbanization rate in 2010 reached about 50 percent from 10.6 percent in 1949, showing a significant urbanization development pro-

 $^{^{24}}$ The UN report predicts the total population of China is 1.449 billion in 2025, and 1.385 billion in 2050, and the population in 2013 is 1.386 billion.



Figure 8: Change in energy mix in different policy scenarios

cess over time.²⁵ It is predicted to further climb to near 60% in 2020 and around 66% in 2030.²⁶ We explore the long-run effects of urbanization on sectoral growth in this section.

Depending on the region where one lives, people have different consumption bundle preferences, which are reflected by parameters in consumption equations.²⁷ The urbanization rate is exogenously given in the model for simplicity.²⁸ We assume the urbanization ratio increases to 60 percent in 2020 and continues to rise up to 66% in 2030 (hereafter URB). The rate of urbanization in 2050 will reach 78 percent, converging to current level of US.

When people move from rural to urban regions as predicted, total rural consumption growth declines from 4 percent to 1.10 percent in the reference case, while total growth of consumption in the cities increases to 4.36 percent. Data in

²⁵According to the newly published 2009 City Development Report of China, an annual report conducted by China's Association of Mayors says that nearly 621.86 million people lived in cities in 2009. The number of cities grew from 132 at the beginning of 1949 to 655 by 2009.

²⁶The world average urbanization ratio in developed countries 85%, and China's urbanization still lags behind the industrial development, which leaves huge room for further development.

²⁷Explicitly, ζ in equation 13, β_{ne} in equation 14, ι in equation 15, and α_{coa} , α_{oil} , α_{gas} in equation 16 are distinguished between regions according to Input-output table data calibration.

²⁸The government is expected to be careful when allocating fiscal spending as it carries out the new urbanization plans. One precondition for urbanization should be ensuring a sufficient and stable supply of agricultural produce, which will require improved efficiency in agricultural production, based on advanced technology and management. In addition, the provision of housing, social security and education for migrant workers and their children once they settle in the cities, will also present problems that must be solved during the urbanization process.

the base year show that people living in cities consume more than rural residents. Hence, welfare of the whole economy increases with the urbanization process (by 0.1 percent).

On the sector level, the agricultural sector (agr) shrinks relative to the reference case with an average growth rate of 3.78 percent. The construction sector (con) benefits from urbanization with an average growth rate of 4.17 percent. It is followed by the water supply industry (wat), with an average growth rate of 4.15 percent, and sectors which are important for city consumers grow, for instance, machinery (mch) and public services (gse). All four energy sectors (ele,coa, oil, gas) grow faster compared to the reference scenario, showing that city residents consume more energy goods or energy intensive goods compared to rural household. The increase in gas is higher than that of other energy sources.



Figure 9: Growth rates of sectors with urbanization development

Table 6 provides the results when carbon policies (CER524, CER361 and ZERO) are implemented in a growing economy with urbanization. As expected, the growth rate of consumption declines and the aggregate welfare loss rises when more stringent climate policies are implemented. Welfare losses in ZERO, CER524, CER361 are 2.68%, 3.24% and 8.62%, respectively. The welfare losses are slightly higher compared to the scenarios without consideration of urbanization. This is due to the fact that urbanization increases the demand for energy

goods or energy intensive products.

Scenario	Average emission per	Urban consumption	Rural consumption	Welfare
	capita per year	growth rate	growth rate	change
URB	No carbon policy	4.36%	1.10%	+0.06%
ZERO	5.4t	4.16%	0.97%	-2.68%
CER524	5.24t	4.13%	0.95%	-3.24%
CER361	3.61t	3.87%	0.73%	-8.62%

Table 6: Consumption growth rate and welfare loss when climate policies are imposed

Sector	No urbanization			Wit	ation		
	ZERO	CER524	CER361	ZERO	CER524	CER361	
agr	4.15	4.17	4.17	3.92	3.94	3.93	
\min	1.54	1.16	-0.56	1.59	1.21	-0.49	
chm	3.14	3.02	2.46	3.13	3.00	2.44	
mch	2.92	2.78	2.18	2.98	2.84	2.24	
oin	4.11	4.15	4.27	4.11	4.14	4.26	
con	3.41	3.32	2.95	3.59	3.51	3.14	
trn	3.13	3.01	2.44	3.15	3.02	2.45	
bnk	3.64	3.60	3.42	3.65	3.62	3.43	
pse	3.95	3.96	3.93	3.91	3.91	3.88	
gse	3.93	3.91	3.70	3.98	3.96	3.74	
rea	4.09	4.10	4.04	4.10	4.11	4.05	
ele	-0.55	-1.00	-2.68	-0.56	-1.02	-2.69	
coa	-1.22	-1.83	-4.15	-1.25	-1.86	-4.19	
oil	0.74	0.29	-1.47	0.76	0.31	-1.44	
gas	2.05	1.72	0.39	2.12	1.80	0.47	
wat	2.81	2.67	2.08	2.92	2.77	2.16	

Table 7: Growth rate of sectors in the long run (in %)

Sectoral diversification follows the similar patterns as described in the last section. Energy intensive sectors decrease relatively more while labor capital intensive sectors are able to adjust and alleviate the effects of climate policies. When urbanization is taking place, sectoral growth changes slightly. As shown in Table 7, sectors which produce goods that are more demanded by urban household, such as *mch, con, gse, wat*, grow at a higher rate, while the growth rates of other sectors (e.g. *agr*) decline.

6 Conclusions

Using a multisector endogenous growth model, the paper derives the costs of carbon policies in China. We argue that growth dynamics constitute the crucial model element permitting reliable calculation of the effects. Intersectoral linkages and spillover effects are also important drivers of macroeconomic development. Capturing the energy sector, with energy as an essential input to production in different sectors, in an accurate way is crucial for the results. More detailed modeling of the interaction between energy input and economic growth results in a more precise estimation of the cost of climate policies.

Our estimation results show that it is significantly easier for a growing economy to achieve stringent emission intensity reduction targets than absolute emission cuts. The welfare loss of achieving 40-45 percent emission intensity reduction in 2020 relative to the 2005 level is less than a half percent. Increasing the stringency of absolute emission targets and including a longer time horizon increases the cost of policies significantly. Welfare cost increase up to 8.3 percent depending on the stringency of the policy, if we assume the same kind of technical progress for the energy sectors as for the other sectors, constant energy prices in the reference case without policy, and a regular discount rate. This reveals that, even taking into account the ability of an economy to innovate and invest according to changing energy market conditions, costs of carbon policies cannot be disregarded when the reference growth rate is high. Of course, accelerated technology development in the energy sector, intensified learning effects, and increasing energy scarcities alleviate the costs of the climate policies. However, increasing urbanization acts in the opposite direction. The sectoral analysis reveals further interesting results. Central sectors in manufacturing such as machinery as well as electricity production have a very high carbon content in international comparison. Accordingly, increased investments to raise productivity associated with one unit of carbon emission have a very high return in the case of China, helping to decrease the cost of climate policies.

The overall assessment of climate policies in China has to include the benefits of reduced temperature rise, which is not treated in this paper. It would involve including important issues such as uncertainty, tipping points, and time lags in the carbon cycle. Nevertheless, we base our policy targets for the long run on an internationally shared carbon budget which appears to be within a realistic range. Provided that the net benefit of climate policy on a global scale is strongly positive, this suggests that also for a large country like China, climate policy is beneficial, provided it is based on a broad international agreement.

There are various possible extensions of our model. Various provinces in China

are very different in terms of income level, energy use, and economic structures. Hence, a multi-region setup would be helpful, to include provincial differences which are especially important when policies across regions are different. Moreover, an endogenous mechanism for the determination of urbanization and ruralurban migration could be included for further analysis. Extending the modeling of the electricity sector to include a comprehensive bottom-up model part for various generation technologies and transmission grid network could also be useful for energy studies.

References

- Acemoglu, Daron, Philippe Aghion, Leonardo Bursztyn, and David Hemous (2012): The Environment and Directed Technical Change. *American Economic Review*, 102(1): 131-66.
- [2] Allcott, Hunt, and Michael Greenstone (2012): Is There an Energy Efficiency Gap? Journal of Economic Perspectives, 26(1): 3-28.
- [3] Bai, Chong-en, Chang-tai Hsieh and Yingyi Qian (2006): The Return to Capital in China. Brookings Papers on Economic Activity, 2006(2), 61-88.
- [4] Blanford, J., R.G. Richels, and T.F. Rutherford (2008): Revised Emissions Growth Projections for China: Why Post-Kyoto Climate Policy Must Look East. Cambridge, Mass.: Harvard Project on International Climate Agreements.
- [5] Bretschger, Lucas, Roger Ramer and Florentine Schwark(2011): Growth effects of carbon policies: applying a fully dynamic CGE model with heterogeneous capital. *Resource and Energy Economics* 33 (4), pp. 963-980.
- [6] Bretschger, Lucas and Sjak Smulders (2012): Sustainability and Substitution of Exhaustible Natural Resources: How Resource Prices Affect Long-Term R&D-Investments, *Journal of Economic Dynamics and Control*, 36 (4): 536-549.
- Bretschger, Lucas (2013): Climate policy and equity principles: fair burden sharing in a dynamic world. *Environment and Development Economics*, 18: 517–536.
- [8] Cullen, Joseph (2013): Measuring the Environmental Benefits of Wind-Generated Electricity. American Economic Journal: Economic Policy, 5(4): 107-33.
- [9] Dai, H., Masui, T., Matsuoka, Y., Fujimori, S. (2011): Assessment of China's climate commitment and non-fossil energy plan towards 2020 using hybrid AIM/CGE model, *Energy Policy* 39, 2875-2887.

- [10] Dixit, A.K. and Stiglitz, J.E. (1977): Monopolistic competition and optimum production diversity. *American Economic Review*, Vol. 67, No.3, 297-308.
- [11] Donnelly, W. A., Johnson, K., Tsigas, M. E., Ingersoll, D. L. (2004): Revised Armington elasticities of substitution, USITC Model and the concordance for constructing consistent set for the GTAP model, USITC Office of Economics Research Note No. 20001-A.
- [12] Ethier, W. J. (1982): National and international returns to scale in the modern theory of international trade. *American Economic Review*, Vol. 72, No. 3, 389-405.
- [13] Fisher-Vanden, K. and Ho, S. (2007): How do market reforms affect China's responsiveness to environment policy? *Journal of Development Economics* 82, 200-233.
- [14] Fleisher, B., Li, H., Zhao, M.Q. (2010): Human capital, economic growth, and regional inequality in China, *Journal of Development Economics*, 92, 215-231.
- [15] Gans, Joshua S. (2012): Innovation and Climate Change Policy. American Economic Journal: Economic Policy, 4(4): 125-45.
- [16] Garbaccio, F., Ho, S. and Jorgenson, W. (1999): Controlling carbon emission in China, *Environment and Development Economics*, Issue 04, 493-518.
- [17] Hasanov, F. (2007): Housing, household portfolio, and intertemporal elasticity of substitution: evidence from the Consumer Expenditure Survey, Macroeconomics 0510011, EconWPA.
- [18] Heggedal, T.R. and Jacobsen, K. (2011): Timing of innovation policies when carbon emissions are restricted: an applied general equilibrium analysis. *Resource and Energy Economics*, Volume 33, Issue 4, 913-937.
- [19] Hotelling, H. (1931): The economics of exhaustible resources, Journal of Political Economy.
- [20] Huebler, M. (2011): Technology diffusion under contraction and convergence: A CGE analysis of China, *Energy Economics* 33, 131-142.

- [21] Jorgenson, D.W, R.J. Goettle, M.S. Ho, and P.J. Wilcoxen (2013): Energy, the Environment and US Economic Growth, in: Handbook of CGE Modeling, Vol. 1, 477-552.
- [22] Kuemmel R., Henn, J., Lindenberger, D. (2002): Capital, labor, energy and creativity: modeling innovation diffusion, *Structural Change and Economic Dynamics* 13, 415-433.
- [23] Liang, Q., Fan, Y., and Wei, Y. (2007): Carbon taxation policy in China: How to protect energy- and trade- intensive sectors? *Journal of Policy Modeling* 29, 311-333.
- [24] Lu, C., Zhou, D. (2009): Industrial energy substitution and a revised Allen elasticity in China, Front. Econ. China, 4(1): 110-124.
- [25] Okagawa, A., Ban, K. (2008): Estimation of substitution elasticities for CGE models, Graduate School of Economics and Osaka School of International Public Policy, Discussion Paper 08-16.
- [26] Popp, David (2002): Induced Innovation and Energy Prices. American Economic Review, 92(1): 160-180.
- [27] Ramer, R. (2011): Economic effects of energy policies: Applying a CGE model with endogenous growth. Diss. ETH No. 19690.
- [28] Romer, P.M. (1990): Endogenous technical change. The Journal of Political Economy, 98 (5) (1990), pp. 71-102.
- [29] Rutherford, T. (2004): Dynamic general equilibrium with GAMS/MPSGE. Lecture Notes for the UNSW Workshop. http://www.mpsge.org/dynamics. pdf
- [30] United Nation (2013): World population prospects the 2012 revision. New York.
- [31] van der Werf, E. (2007): Production functions for climate policy modeling: An empirical analysis, *Energy Economics* 30, pp. 2964-2979.

- [32] Vennemo, H., Aunan, K., He, J., Hu, T., Li, S. (2009): Benefits and costs to China of three different climate treaties, *Resource and Energy Economics* 31, 139-160.
- [33] Wang, K., Wang, C., and Chen, J. (2009): Analysis of the economic impact of different Chinese climate policy options based on a CGE model incorporating endogenous technological change, *Energy Policy*, Volume 37, Issue 8, 2930-2940.
- [34] Wolfram, Catherine, Orie Shelef, and Paul Gertler (2012): How Will Energy Demand Develop in the Developing World? Journal of Economic Perspectives, 26(1): 119-38.
- [35] Song, Zheng, Kjetil Storesletten, and Fabrizio Zilibotti (2011): Growing Like China. American Economic Review, 101(1): 196-233.
- [36] Zhang, Z.X. (1998): Macroeconomic effects of CO2 emission limits: a computable general equilibrium analysis for China, *Journal of Policy Modeling*, 20(2), 213-250.
- [37] Zhu, Xiaodong (2012): Understanding China's Growth: Past, Present, and Future. Journal of Economic Perspectives, 26(4): 103-24.
- [38] Zheng, Siqi, and Matthew E. Kahn (2013): Understanding China's Urban Pollution Dynamics. *Journal of Economic Literature*, 51(3): 731-72.

7 Appendix: sensitivity analysis on substitution elasticities

This section shows some sensitivity analysis on elasticity of substitutions used in the numerical model. We vary the value of Armington trade elasticities η and elasticity of substitution between energy sources in intermediate production. The results are shown in Table 8.

Scenarios	W	elfare chan	ge	Aggregate	consumpti	on growth
Trade elasticity η						
	$\eta - 1$	η	$\eta + 1$	$\eta - 1$	η	$\eta + 1$
CER524	-3.20%	-3.10%	-3.00%	3.78%	3.79%	3.79%
CER361	-8.70%	-8.33%	-8.00%	3.52%	3.54%	3.55%
ZERO	-2.60%	-2.57%	-2.50%	3.81%	3.82%	3.82%
Energy Substitution σ_{egy}						
	$\sigma_{egy} = 0.1$	$\sigma_{egy} = 0.8$	$\sigma_{egy} = 1.5$	$\sigma_{egy} = 0.1$	$\sigma_{egy} = 0.8$	$\sigma_{egy} = 1.5$
CER524	-3.11%	-3.10%	-3.09%	3.78%	3.79%	3.79%
CER361	-8.38%	-8.33%	-8.29%	3.54%	3.54%	3.54%
ZERO	-2.57%	-2.57%	-2.57%	3.82%	3.82%	3.82%

Table 8: Aggregate effects when varying elasticity of substitutions

As we expected, lowering trade elasticities makes it difficult to substitute between domestically produced goods and imported goods. That is, household consumption is more domestic dependent compared to higher values. As the prices of goods produced in home country are now relatively expensive because carbon policies increases the input price of energy, people now can buy less goods with the same amount of money. Moreover, consumers are not able to buy more imported goods as a substitute as the deline in trade elasticities. Hence, lowering trade elasticities results in higher welfare loss and lower consumption growth.

Energy substitution elasticity affects the intermediate production, and hence the final output. Lower value means all energy sources are not substitutable, while higher value suggests easy substitution between sources. It is obvious that higher substitution elasticitity gives firms more flexibility in adapting to price change due to carbon policies, making it less costly to implementing emission reduction polices. On the contrary, lower substitution elasticity indicates the rigidity in changing production inputs, expensive sources are still heavily required for production. This leads to higher cost to implement carbon policy.

However, all these robustness check suggests that our cost estimation are stable

in magnitude. The results are not very sensitive to the elasticities. Of course, we can run analysis for all other elasticities. As from our experience, the above two are the most relevant to this paper.

Working Papers of the Center of Economic Research at ETH Zurich

(PDF-files of the Working Papers can be downloaded at www.cer.ethz.ch/research).

- 14/201 L. Bretschger and L. Zhang Carbon policy in a high-growth economy: The case of China
- 14/200 N. Boogen, S. Datta and M. Filippini Going beyond tradition: Estimating residential electricity demand using an appliance index and energy services
- 14/199 V. Britz and H. Gersbach Experimentation in Democratic Mechanisms
- 14/198 M. Filippini and E. Tosetti Stochastic Frontier Models for Long Panel Data Sets: Measurement of the Underlying Energy Efficiency for the OECD Countries
- 14/197 M. Filippini and W. Greene Persistent and Transient Productive Inefficiency: A Maximum Simulated Likelihood Approach
- 14/196 V. Britz, P. J.-J. Herings and A. Predtetchinski Equilibrium Delay and Non-existence of Equilibrium in Unanimity Bargaining Games
- 14/195 H. Gersbach, M. T. Schneider and O. Tejada Coalition-Preclusion Contracts and Moderate Policies
- 14/194 A. Bommier Mortality Decline, Impatience and Aggregate Wealth Accumulation with Risk-Sensitive Preferences
- 14/193 D. Harenberg and A. Ludwig Social Security and the Interactions Between Aggregate and Idiosyncratic Risk
- 14/192 W. Mimra, A. Rasch and C. Waibel Second Opinions in Markets for Expert Services: Experimental Evidence
- 14/191 G. Meunier and J-P. Nicolai Higher Costs for Higher Profits: A General Assessment and an Application to Environmental Regulations
- 14/190 A. Alberini, M. Bareit and M. FilippiniDoes the Swiss Car Market Reward Fuel Efficient Cars? Evidence from HedonicPricing Regressions, Matching and a Regression Discontinuity Design
- 14/189 J-P. Nicolai and J. Zamorano "Windfall profits 2.0" during the third phase of the EU-ETS

- 13/188 S. Hector Accounting for Different Uncertainties: Implications for Climate Investments
- 13/187 J-P. Nicolai Ex-Ante vs. Ex-Post Pollution Permit Regulation under Imperfect Competition
- 13/186 C. Christin, J-P. Nicolai and J. Pouyet Pollution Permits, Imperfect Competition and Abatement Technologies
- 13/185 L. Bretschger and S. Valente International Resource Tax Policies Beyond Rent Extraction
- 13/184 L. Bretschger and C. Karydas Optimum Growth and Carbon Policies with Lags in the Climate System
- 13/183 M. Filippini and L. Zhang Measurement of the "Underlying energy efficiency" in Chinese provinces
- 13/182 J. G. Becker and H. Gersbach A Theory of Threshold Contracts
- 13/181 M. Filippini and L. C. Hunt 'Underlying Energy Efficiency' in the US
- 13/180 O. Tejada Complements and Substitutes in Generalized Multisided Assignment Economies
- 13/179 A. Rasch and C. Waibel What drives fraud in a credence goods market? Evidence from a field study
- 13/178 B. Lanz and S. Rausch Cap-and-Trade Climate Policy, Free Allowances, and Price-Regulated Firms
- 13/177 H. Karle Creating Attachment through Advertising: Loss Aversion and Pre-Purchase Information
- 13/176 W. Mimra, A. Rasch and C. Waibel Price competition and reputation in credence goods markets: Experimental evidence
- 13/175 H. Gersbach and M. T. Schneider On the Global Supply of Basic Research
- 13/174 B. Lanz, T. F. Rutherford and J. E. Tilton Subglobal climate agreements and energy-intensive activities: An evaluation of carbon leakage in the copper industry
- 13/173 L. Blazquez, F. Heimsch and M. Filippini The Economic Crisis and Residential Electricity Consumption in Spanish Provinces: A Spatial Econometric Analysis

- 13/172 A. Bommier and F. Le Grand A Robust Approach to Risk Aversion
- 13/171 P. J. Agrell, M. Farsi, M. Filippini and M. Koller Unobserved heterogeneous effects in the cost efficiency analysis of electricity distribution systems
- 13/170 S. Andrade de Sa , C. Palmer and S. Di Falco Dynamics of Indirect Land-Use Change: Empirical Evidence from Brazil
- 12/169 L. Leinert Does the Oil Price Adjust Optimally to Oil Field Discoveries?
- 12/168 F. Lechthaler and L. Leinert Moody Oil - What is Driving the Crude Oil Price?
- 12/167 L. Bretschger, R. Ramer and L. Zhang Economic effects of a nuclear phase-out policy: A CGE analysis
- 12/166 L. Bretschger and N. Suphaphiphat Use Less, Pay More: Can Climate Policy Address the Unfortunate Event for Being Poor?
- 12/165 P. S. Schmidt and T. Werner Verified emissions and stock prices: Is there a link? - An empirical analysis of the European Emission Trading Scheme
- 12/164 H. Gersbach, S. Imhof and O. Tejada Channeling the Final Say in Politics
- 12/163 J. Daubanes and L. Leinert Optimum Tariffs and Exhaustible Resources: Theory and Evidence for Gasoline
- 12/162 C. N. Brunnschweiler, C. Jennings and I. A. MacKenzie Rebellion against Reason? A Study of Expressive Choice and Strikes
- 12/161 A. Lange and A. Ziegler Offsetting versus Mitigation Activities to Reduce CO2 Emissions: A Theoretical and Empirical Analysis for the U.S. and Germany
- 12/160 L. Bretschger and F. Lechthaler Common Risk Factors and the Macroeconomy: New Evidence from the Japanese Stock Market
- 12/159 H. Gersbach and Q. Oberpriller Rules vs. Targets: Climate Treaties under Uncertainty